\begin{table}%t2
     \caption{\label{properties}Summary of galaxy properties.}
    
     \begin{tabular}{p{0.18\linewidth}cccccc}
       \hline
       \hline
       \noalign{\smallskip}
                                    & {{\rm NGC~45}}  &  {{\rm NGC~1313}}  & {{\rm NGC~4395}} & {{\rm NGC~5236}} & {{\rm NGC~7793}} \\
       \noalign{\smallskip}
       \hline
       \noalign{\smallskip}
       Type$^{{a}}$          &{\rm SA(s)dm}&  {\rm SB(s)d} & {\rm SA(s)m}  & {\rm SAB(s)c}  & {\rm SA(s)d}          \\  
       B$^{\dag}$	            & $11.37\pm0.11$   &  $9.66  \pm  0.41$ & $10.84  \pm  0.26$ & $7.82  \pm  0.21$   & $9.71  \pm  0.15$   \\
       $B-V^{\dag}$                 & 0.71           &  0.49            &    0.46          & 0.66              &  0.54               	\\
       $A_{B}^{{\rm \ddag}}$     & 0.09	     &  0.47            &    0.08          & 0.29              & 0.08            	\\
       (m-M)$_0$                    & $28.42\pm 0.41^1$&  $28.2^2$          &    $28.1^3$        &    $27.84\pm0.15^4$ &    $27.6\pm0.2^5$           \\
       ${[{\rm Fe}/{\rm H}]}$ &-               &-                 & -                &-                  & $-1.22^{6}$\\
       ${[12 + \log ({\rm O}/{\rm H})]}$&-     & -                &$8.48 \pm0.13^{7}$  & $8.9-9.1^{8}$       & - \\
       $Z$                          &-               &$0.008^{9}$         &-                 &-                  &-   \\
       $T_N ^{{\rm 10}}$         &0.28            &1.12              &0.21              &1.77               &1.21\\         
       $T_L(U) ^{{\rm 11}}$      & 0.24           &1.47              &0.07              &2.23               &1.15\\	
       \hline		
     \end{tabular}											\par 
$^{{a}}$ Retrieved from the NASA/IPAC Extragalactic Database (NED);
$^{\dag}$ \citet{paturel} retrieved from Hyperleda;
$^{{\rm \ddag}}$ \citet{schlegel} retrieved from NED; 
$^1$ \citet{Bottinelli}; 
$^2$ \citet{deVaucouleurs63};
$^3$ \citet{karachentsev98}; 
$^4$ \citet{deVaucouleurs79a}; 
$^5$ \citet{Carignan85}; 
$^6$ \citet{karachentsev2003}; 
$^7$ \citet{vanZee98};
$^8$ \citet{Bresolin2002};
$^9$ \citet{Soeren07}.
$^{10}$ \citet{lr99}.
$^{11}$ \citet{SoerenTom2000}.
\normalsize
     \end{table}